Author's response to reviews

Title: British pain clinic practitioners' recognition and use of the bio-psychosocial pain management model for patients when physical interventions are ineffective or inappropriate: results of a qualitative study.

Authors:

Geoffrey Harding (geoffrey.harding@pms.ac.uk)
John Campbell (John.campbell@pms.ac.uk)
Suzanne Parsons (suzanne.parsons@PickerEurope.ac.uk)
Anisur Rahman (anisur.rahman@ucl.ac.uk)
Martin Underwood (m.underwood@warwick.ac.uk)

Version: 4 Date: 11 March 2010

Author's response to reviews: see over
We have addressed both major and minor revisions requested. Our response to these are highlighted below.

Major revisions (we require the author to make these changes)

Competing interests: Please include a `Competing interests` section between the Conclusions and Authors' contributions.

This section has now been added

Authors' contributions: Please place an Authors' contributions section towards the end of the manuscript text, directly preceding the Acknowledgements.

This section has now been added

Please state that all authors read and approved the final manuscript.

This statement has now been added

Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references.

A tables section has now been added following the references.

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].`

We have replaced the superscript numbers and placed them in square brackets

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Email address: please include email addresses for all the authors on the title page

These have now been added in the required format.
Background: Please include the heading 'Background' above the text after the abstract.

This has now been added

Table placement text: Please remove this.

Removed

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Removed

Table: Please refer to this as 'Table 1', rather than 'Table one'.

Done